\def\CTeXPreproc{Created by ctex v0.2.12, don't edit!}\documentclass{article}
\usepackage{latexsym,amsmath,amsthm,amssymb,cite}
%\usepackage[notcite,notref]{showkeys}
\headsep5mm\topmargin-5mm\oddsidemargin0pt\evensidemargin0pt
\setlength{\textwidth}{150mm}\setlength{\textheight}{240mm}
\setlength{\parskip}{3pt plus1pt
minus2pt}\setlength{\baselineskip}{20pt plus2pt minus1pt}

\numberwithin{equation}{section}
\newtheorem{theorem}{\indent Theorem}[section]\newtheorem{corollary}{\indent Corollary}[section]
\newtheorem{remark}{\indent Remark}[section]\newtheorem{lemma}{\indent Lemma}[section]
\newtheorem{example}{\indent Example}[section]
\newcommand{\beq}{\begin{equation}}
\newcommand{\eeq}{\end{equation}}

\title{Necessary and sufficient conditions for oscillation of second-order differential equation with several delays}
\author{{Shyam Sundar Santra}\\
\small\textit{Department of Mathematics, JIS College of Engineering, Kalyani - 741235, India}\\
\small\textit{{E}-mail: shyam01.math@gmail.com}}

\begin{document}
\begin{thebibliography}{99}

\bibitem{Agarwal1}
R. P. Agarwal, M. Bohner, T. Li, C. Zhang; {\it Oscillation of second-order differential equations with a sublinear neutral term,} Carpathian J. Math.,  {\bf 30} (2014), 1--6.

\bibitem{Agarwal2}
R. P. Agarwal, M. Bohner, T. Li, C. Zhang; {\it Oscillation of second-order Emden–Fowler neutral delay differential equations,} Ann. Mat. Pura Appl. {\bf 193}(4),  (2014), 1861--1875.

\bibitem{Agarwal3}
R. P. Agarwal, M. Bohner, T. Li, C. Zhang; {\it Even-order half-linear advanced differential equations: improved criteria in oscillatory and asymptotic properties,} Appl. Math. Comput., {\bf 266} (2015), 481--490.

\bibitem{Agarwal4}
R. P. Agarwal, C. Zhang, T. Li; \emph{Some remarks on oscillation of second order neutral differential equations}, Appl. Math. Comput., {\bf 274} (2016), 178--181.

\bibitem{Brands}
J. J. M. S. Brands; {\it Oscillation theorems for second-order functional-differential equations}, J. Math. Anal. Appl., {\bf 63}(1): (1978), 54-–64.

\bibitem{Baculikova2}
B. Baculikova, J. Dzurina; {\it Oscillation theorems for second-order neutral differential equations}, Comput. Math. Appl., {\bf 61} (2011), 94--99.

\bibitem{Baculikova3}
B. Baculikova, J. Dzurina; {\it Oscillation theorems for second-order nonlinear neutral differential equations}, Comput. Math. Appl., {\bf 62} (2011), 4472--4478.

\bibitem{Baculikova4}
B. Baculikova, T. Li, J. Dzurina; {\it Oscillation theorems for second order neutral differential equations}, Electron. J. Qual. Theory Differ. Equ., {\bf 74} (2011), 1--13.

\bibitem{boh}
M. Bohner, S. R. Grace,  I. Jadlovska; \emph{Oscillation criteria for second-order neutral delay differential equations}, Electron. J. Qual. Theory Differ. Equ., (2017), 1 --12.

\bibitem{chat1}
G.  E. Chatzarakis, J. Dzurina, I. Jadlovska; \emph{New oscillation criteria for second-order half-linear advanced differential equations}, Appl. Math. Comput., {\bf 347} (2019), 404--416.

\bibitem{chat2}
G. E. Chatzarakis, I. Jadlovska; \emph{Improved oscillation results for second-order half-linear delay differential equations}, Hacet. J. Math. Stat., {\bf 48}(1): (2019), 170–179.

\bibitem{chat3}
G. E. Chatzarakis, S. R. Grace, I. Jadlovska, T. Li, E. Tunc; \emph{Oscillation criteria for third order Emden–Fowler differential equations with unbounded neutral coefficients}, Complexity, 2019 (2019), 1--7.

\bibitem{dzurina} J.~D\v{z}urina; {\it {O}scillation theorems for second-order advanced neutral differential equations,} Tatra Mt. Math. Publ., {\bf 48} (2011), 61--71.

\bibitem{dzurina1} J. ~D\v{z}urina, S. R. Grace, I. Jadlovska, T. Li; \emph{Oscillation criteria for second-order Emden-Fowler delay differential equations with a sublinear neutral term,} Math. Nachr., (2019) in press.

\bibitem{Fisnarova}
S. Fisnarova, R. Marik; \emph{Oscillation of neutral second-order half-linear differential equations without commutativity in delays}, Math. Slovaca, {\bf 67}(3), (2017), 701--718.

\bibitem{Grace}
S. R. Grace, J. Džurina,  I. Jadlovska, T. Li; {\it An improved approach for studying oscillation of second-order neutral delay differential equations}, J. Inequ. Appl.,  (2018), 11 pages.

\bibitem{zbMATH03547704} J.~Hale; {\it {T}heory of {F}unctional {D}ifferential {E}quations.} Applied Mathematical Sciences, 2nd ed. {\bf 3}. New York - Heidelberg - Berlin: Springer-Verlag, 1977. 1.

\bibitem{karpuz}
B. Karpuz, S. S. Santra; {\it Oscillation theorems for second-order nonlinear delay differential equations of neutral type,} Hacet. J. Math. Stat., {\bf 48}(3): (2019), 633--643.

\bibitem{Li7}
T. Li, Y. V. Rogovchenko; {\it Oscillation theorems for second-order nonlinear neutral delay differential eqquations,} Abstr. Appl. Anal., {\bf 2014}  (2014), ID 594190, 1--11.

\bibitem{Li8}
T. Li, Y. V. Rogovchenko; \emph{Oscillation of second-order neutral differential equations,} Math. Nachr., {\bf 288} (2015), 1150--1162.

\bibitem{Li9}
T. Li, Y. V. Rogovchenko; \emph{Oscillation criteria for second-order superlinear Emden-Fowler neutral differential equations,} Monatsh. Math., {\bf 184} (2017), 489--500.

\bibitem{li1}
Q. Li, R. Wang, F. Chen, T. Li; \emph{Oscillation of second-order nonlinear delay differential equations with nonpositive neutral coefficients,} Adv. Difference Equations, (2015), 7 pages.

\bibitem{li2}
H. Li, Y. Zhao, Z. Han; \emph{New oscillation criterion for Emden–Fowler type nonlinear neutral delay differential equations,} J. Appl. Math. Comput., {\bf 60}(1-2), (2019), 191--200.

\bibitem{pinelas}
S. Pinelas, S. S. Santra; \emph{Necessary and sufficient condition for oscillation of nonlinear neutral first-order differential equations with several delays,} J. Fixed Point Theory Appl., {\bf 20}(27): (2018), 1--13.

\bibitem{Qian8}
Y. Qian, R. Xu; {\it Some new osciilation criteria for higher order quasi-linear neutral delay differential equations,} Differ. Equ. Appl., {\bf 3} (2011), 323--335.

\bibitem{santra1}
S. S. Santra; {\it Existence of positive solution and new oscillation criteria for nonlinear first-order neutral delay differential equations,} Differ. Equ. Appl., {\bf 8}(1): (2016), 33--51.

\bibitem{santra2}
S. S. Santra; {\it Oscillation analysis for nonlinear neutral differential equations of second-order with several delays,} Mathematica, {\bf 59(82)}(1-2): (2017), 111--123.

\bibitem{santra4}
S. S. Santra; \emph{Oscillation analysis for nonlinear neutral differential equations of second-order with several delays and forcing term,} Mathematica, {\bf 61}(84)(1): (2019), 63--78.

\bibitem{shyam6}
S. S. Santra; \emph{Necessary and sufficient condition for the solutions of first-order neutral differential equations to be oscillatory or tend to zero,} KYUNGPOOK Math. J., {\bf 59} (2019), 73-82.

\bibitem{santra5}
S. S. Santra; \emph{Necessary and sufficient condition for oscillatory and asymptotic behaviour of second-order functional differential equations,} Krag. J. Math., {\bf 44}(3): (2020), 459--473.

\bibitem{akt}
A. K. Tripathy, B. Panda, A. K. Sethi; \emph{On oscillatory nonlinear second-order neutral delay differential equations,} Differ. Equ. Appl., {\bf 8}(2): (2016), 247--258.

\bibitem{Wong} J. S. W. Wong; {\it Necessary and suffcient conditions for oscillation of second-order neutral differential equations,} J. Math. Anal. Appl., \textbf{252}(1): (2000), 342--352.

\bibitem{Yang10}
Q. Yang, Z. Xu; {\it Oscillation criteria for second-order quasi-linear neutral delay differential equations on time scales,} Comput. Math. Appl., {\bf 62} (2011), 3682--3691.

\bibitem{Ye11}
L. Ye, Z. Xu; {\it Oscillation criteria for second-order quasi-linear neutral delay differential equations,} Appl. Math. Comput., {\bf 207} (2009), 388--396.

\bibitem{Zhang} C. Zhang, R. P. Agarwal, M. Bohner, T. Li; \emph{Oscillation of second-order nonlinear neutral dynamic equations with noncanonical operators,} Bull. Malays. Math. Sci. Soc., {\bf 38} (2015), 761–778.
\end{thebibliography}

\date{}
\maketitle
\begin{abstract}
In this paper, necessary and sufficient conditions are establish of the solutions to second-order delay differential equations of the form
\begin{equation}
\Big(r(t)\big(x'(t)\big)^\gamma\Big)' +\sum_{i=1}^m q_i(t)f_i\big(x(\sigma_i(t))\big)=0 \text{ for } t \geq t_0,\notag
\end{equation}
We consider two cases when $f_i(u)/u^\beta$ is non-increasing for $\beta<\gamma$, and non-decreasing for $\beta>\gamma$ where $\beta$ and $\gamma$ are the quotient of two positive odd integers. Our main tool is Lebesgue's Dominated Convergence theorem. Examples illustrating the applicability of the results are also given, and state an open problem.

\noindent\textbf{Keywords}: Oscillation; nonoscillation; nonlinear;  delay argument; second-order differential equation; Lebesgue's Dominated Convergence theorem.

\noindent\textbf{Mathematics Subject Classification 2010}: 34C10, 34C15, 34K11.
\end{abstract}


\section{Introduction}

In this article we consider the differential equation
\begin{gather} \label{e1}
\Big(r(t)\big(x'(t)\big)^\gamma\Big)'
 +\sum_{i=1}^m q_i(t)f_i\big(x(\sigma_i(t))\big)=0, \quad\text{for } t \geq t_0,
\end{gather}
where $\gamma$ is the quotient of two positive odd integers, and the functions $f_i,p, q_i,r,\sigma_i$ are continuous
that satisfy the conditions stated below;
\begin{itemize}
\item[(A1)] $\sigma_i\in C([0,\infty),\mathbb{R})$,
 $\sigma_i(t)<t$, $\lim_{t\to\infty}\sigma_i(t)=\infty$.

\item[(A2)] $r\in C^1([0,\infty),\mathbb{R})$,
$q_i\in C([0,\infty),\mathbb{R})$; $0<r(t)$,
$0\leq q_i(t)$,  for all $t\geq 0$ and  $i=1,2,\dots,m$;
$\sum q_i(t)$ is not identically zero in any interval $[b,\infty)$.

\item[(A3)] $f_i \in C(\mathbb{R},\mathbb{R})$ is non-decreasing and
$f_i(x)x>0$ for $x\neq 0$, $i=1,2,\dots,m$.

\item[(A4)] $\int_0^\infty r^{-1/\gamma}(\eta)\,d\eta=\infty$; let
$R(t)=\int_0^t r^{-1/\gamma}(\eta)\,d\eta$.

\end{itemize}
The main feature of this article is having  conditions that are both
necessary and sufficient for the oscillation of all solutions to
\eqref{e1}.


In 1978, Brands \cite{Brands} has proved that for bounded delays, the solutions of
\[
x''(t)+q(t)x(t-\sigma(t)) =0
\]
are oscillatory if and only if the solutions of $x''(t)+q(t)x(t) =0$ are oscillatory. In \cite{chat1,chat2} Chatzarakis {\it et al.} have considered a more general second-order half-linear differential equation of the form
\begin{align}\label{chat}
\big(r(x^{\prime})^\alpha\big)^{\prime}(t)+q(t)x^{\alpha}(\sigma(t))=0,
\end{align}
and established new oscillation criteria for \eqref{chat} when $\lim_{t \to \infty}\Pi(t)=\infty$ and $\lim_{t \to \infty}\Pi(t)<\infty$.

Wong \cite{Wong} has obtained the necessary and sufficient conditions for oscillation of solutions of
\[
 \bigl(x(t)+px(t-\tau)\bigr)''+ q(t)f\big(x(t-\sigma)\big)=0, \quad -1<p<0.
\]
in which the neutral coefficient and delays are constants. However, we have seen in \cite{Baculikova2,dzurina} that the authors  Bacul\v{i}kov\v{a} and  D\v{z}urina have studied
\begin{align} \label{e2}
\Big(r(t)\big(z'(t)\big)^\gamma\Big)' + q(t)x^{\alpha}(\sigma(t))=0, \quad z(t)=x(t)+p(t)x(\tau(t)), \quad t\geq t_0,
\end{align}
and established sufficient conditions for  oscillation of solutions of \eqref{e2} using comparison techniques when $\gamma=\alpha=1$, $0\leq{}p(t)<\infty$ and $\lim_{t \to \infty}\Pi(t)=\infty$. In same technique, Baculikova and D\v{z}urina  \cite{Baculikova3} have considered \eqref{e2} and obtained sufficient conditions for oscillation of the solutions of \eqref{e2} by considering the assumptions $0\leq{}p(t)<\infty$ and $\lim_{t \to \infty}\Pi(t)=\infty$. In \cite{akt}, Tripathy {\it et al.} have studied \eqref{e2} and established several sufficient conditions for oscillations of the solutions of \eqref{e2}  by considering the assumptions $\lim_{t \to \infty}\Pi(t)=\infty$ and $\lim_{t \to \infty}\Pi(t)<\infty$ for different ranges of the neutral coefficient $p$. In  \cite{boh}, Bohner {\it et al.} have obtained sufficient conditions for oscillation of solutions of \eqref{e2} when $\gamma=\alpha$, $\lim_{t \to \infty}\Pi(t)<\infty$ and $0 \leq p(t)<1$. Grace {\it et al.} \cite{Grace} have established sufficient conditions for the oscillation of the solutions of \eqref{e2} when $\gamma=\alpha$ and by considering the assumptions $\lim_{t \to \infty}\Pi(t)<\infty$, $\lim_{t \to \infty}\Pi(t)=\infty$ and $0 \leq p(t)<1$. In \cite{li2}, Li {\it et al.} have established sufficient conditions for the oscillation of the solutions of \eqref{e2}, under the assumptions $\lim_{t \to \infty}\Pi(t)<\infty$ and $p(t) \geq 0$. Karpuz and Santra \cite{karpuz} have obtained several sufficient conditions for the oscillatory and asymptotic behavior of the solutions of
 \begin{equation}\notag
\bigl(r(t)(x(t)+p(t)x(\tau(t)))^\prime\bigr)^\prime+q(t)f\bigl(x(\sigma(t))\bigr)=0,
\end{equation}
by considering the assumptions $\lim_{t \to \infty}\Pi(t)<\infty$ and $\lim_{t \to \infty}\Pi(t)=\infty$ for different ranges of $p$.

For further work on the oscillation of the solutions to this type of equations, we refer the readers to \cite{Agarwal1,Agarwal2,Agarwal3,Agarwal4,Baculikova4,chat3,dzurina1,Grace,Li7,Li8,Li9,li1,pinelas,Qian8,santra1,santra2,santra4,Zhang}. Note that the majority of publications consider only sufficient conditions, and and merely a few consider  necessary and sufficient conditions. Hence, the objective in this work is to establish both necessary and sufficient conditions for the oscillatory and asymptotic behavior of solutions of \eqref{e1} without using the comparison and the Riccati techniques.

Delay differential equations have several applications in the natural sciences and engineering. For example, they often appear in the study of distributed networks containing lossless transmission lines (see for e.g. \cite{zbMATH03547704}). In this paper, we restrict our attention to the study \eqref{e1}, which includes the class of functional differential equations of neutral type.


By a solution to equation \eqref{e1}, we mean a function $x\in \mathrm{C}([T_{x},\infty),\mathbb{R})$, where $T_{x}\geq{}t_{0}$,
such that $rx^{\prime}\in\mathrm{C}^{1}([T_{x},\infty),\mathbb{R})$, satisfying \eqref{e1} on the interval $[T_{x},\infty)$.
A solution $x$ of \eqref{e1} is said to be proper if $x$ is not identically zero eventually,
i.e., $\sup\{|x(t)|:\ t\geq T\}>0$ for all $T\geq{}T_{x}$.
We assume that \eqref{e1} possesses such solutions.
A solution of \eqref{e1} is called oscillatory if it has arbitrarily large zeros on $[T_{x},\infty)$;
otherwise, it is said to be non-oscillatory.
\eqref{e1} itself is said to be oscillatory if all of its solutions are oscillatory.

\begin{remark}
When the domain is not specified explicitly,
all functional inequalities considered in this paper are assumed to hold eventually,
i.e., they are satisfied for all $t$ large enough.
\end{remark}



\section{Main Results}

\begin{lemma} \label{lem1}
Assume {\rm (A1)--(A4)}, and that
$x$ is an eventually positive solution of \eqref{e1}. Then there exist $t_1\geq t_0$ and $\delta>0$ such that
\begin{gather}
0<x(t)\leq \delta R(t), \label{e21a}\\
\begin{aligned}
&\big(R(t)-R(t_1)\big)
  \Big[\int_t^\infty \sum_{i=1}^m q_i(\zeta)f_i\big(x(\sigma_i(\zeta))\big)\,d\zeta \Big]^{1/\gamma} \leq x(t)\,,
\end{aligned} \label{e21b}
\end{gather}
for $t\geq t_1$.
\end{lemma}

\begin{proof}
Let $x$ be an eventually positive solution.
Then by (A1) there exists a $t^*$ such that $x(t)>0$, $x(\tau(t))>0$ and
$x(\sigma_i(t))>0$ for all $t\geq t^*$ and $i=1,2,\dots,m$. From \eqref{e1} it follows that
\beq \label{e22}
\begin{gathered}
\Big(r(t)\big(x'(t)\big)^\gamma\Big)'
 = -\sum_{i=1}^m q_i(t)f_i\big(x(\sigma_i(t))\big)\leq 0.
\end{gathered}
\eeq
Therefore, $r(t)\big(x'(t)\big)^\gamma$
 is non-increasing for $t\geq t^*$. Next we show the $r(t)\big(x'(t)\big)^\gamma$ is positive.
By contradiction assume that $r(t)\big(x'(t)\big)^\gamma\leq 0$
at a certain time $t\geq t^*$.
Using that $\sum q_i$ is not identically zero on any interval $[b,\infty)$,
and that $f(x)>0$ for $x>0$, by \eqref{e22}, there exist
 $t_2\geq t^*$ such that
\[
r(t)\big(x'(t)\big)^\gamma
\leq r(t_2)\big(x'(t_2)\big)^\gamma<0\quad\text{for all }t\geq t_2.
\]
Recall that $\gamma$ is the quotient of two positive odd integers.
Then
\[
x'(t)\leq \Big(\frac{r(t_2)}{r(t)}\Big)^{1/\gamma}  x'(t_2)
\quad\text{for } t\geq t_2\,.
\]
Integrating from $t_2$ to $t$, we have
\beq \label{e22b}
 x(t)\leq x(t_2)+ \big(r(t_2)\big)^{1/\gamma} x'(t_2) \big(R(t)-R(t_2)\big)\,.
\eeq
By (A4), the right-hand side approaches $-\infty$; then
$\lim_{t\to \infty} x(t)=-\infty$. This is a contradiction to the fact that $x(t)>0$. Therefore $r(t)\big(x'(t)\big)^\gamma >0$ for all $t\geq t^*$. From $r(t)\big(x'(t)\big)^\gamma$ being non-increasing, we have
\[
x'(t)\leq \Big(\frac{r(t_1)}{r(t)}\Big)^{1/\gamma} x'(t_1)\quad\text{for } t\geq t_1\,.
\]
Integrating this inequality from $t_1$ to $t$, and using that $x$ is continuous,
\[
 x(t)\leq x(t_1)+ \big(r(t_1)\big)^{1/\gamma} x'(t_1) \big(R(t)-R(t_1)\big)\,.
\]
Since $\lim_{t\to\infty}R(t)=\infty$, there exists a positive constant $\delta$
such that \eqref{e21a} holds.

Since $r(t)\big(x'(t)\big)^\gamma$ is positive and non-increasing,
$\lim_{t\to\infty} r(t)\big(x'(t)\big)^\gamma$ exists and is non-negative.
Integrating \eqref{e1} from $t$ to $a$, we have
\[
r(a)\big(x'(a)\big)^\gamma
-r(t)\big(x'(t)\big)^\gamma
+\int_t^\infty \sum_{i=1}^m q_i(\eta)f_i\big(x(\sigma_i(\eta))\big)\,d\eta=0
\]
Computing the limit as $a\to\infty$,
\beq
r(t)\big(x'(t)\big)^\gamma
\geq  \int_t^\infty \sum_{i=1}^m q_i(\eta)f_i\big(x(\sigma_i(\eta))\big)\,d\eta. \label{e23}
\eeq
Then
\[
x'(t) \geq \Big[\frac{1}{r(t)}
\int_t^\infty \sum_{i=1}^m q_i(\eta)f_i\big(x(\sigma_i(\eta))\big)\,d\eta \Big]^{1/\gamma}.
\]
Since $x(t_1)> 0$, integrating the above inequality yields
\[
x(t)
\geq \int_{t_1}^\eta
\Big[\frac{1}{r(\eta)}\int_\eta^\infty \sum_{i=1}^m q_i(\zeta)f_i\big(x(\sigma_i(\zeta))\big)\,d\zeta\Big]^{1/\gamma}\,d\eta
\]
Since the integrand is positive, we can increase the lower limit of
integration from $\eta$ to $t$, and then use the definition of $R(t)$, to obtain
\begin{align*}
x(t) \geq \big(R(t)-R(t_1)\big)
\Big[\int_t^\infty \sum_{i=1}^m q_i(\zeta)f_i\big(x(\sigma_i(\zeta))\big)\,d\zeta
 \Big]^{1/\gamma},
\end{align*}
which yields \eqref{e21b}.
\end{proof}

For the next theorem we assume that there exists a constant $\beta$,
the quotient of two positive odd integers, with $\beta<\gamma$, such that
\beq \label{f/decr}
\frac{f_i(u)}{u^\beta}\text{ is non-increasing for }0<u,\; i=1,2,\dots, m\,.
\eeq
For example  $f_i(u)=|u|^\alpha\operatorname{sgn}(u)$, with $0<\alpha<\beta$
satisfies this condition.
\begin{theorem} \label{thm1}
Under assumptions {\rm (A1)--(A4)} and \eqref{f/decr}, each solution of \eqref{e1} is oscillatory if and only if
\beq \label{e24}
\begin{aligned}
\int_0^\infty \sum_{i=1}^m q_i(\eta)f_i(\delta R(\sigma_i(\eta)))\,d\eta =\infty \quad \forall \delta >0\,.
\end{aligned}
\eeq
\end{theorem}

\begin{proof}
We prove sufficiency by contradiction.
Initially we assume that a solution  $x$ is eventually positive. So, Lemma \ref{lem1} holds, and then there exists $t_1 \geq t_0$ such that
\[
x(t)\geq \big(R(t)-R(t_1)\big) w^{1/\gamma}(t) \geq 0 \quad \forall
t\geq t_1\,,
\]
where
\[
w(t)=\int_t^\infty \sum_{i=1}^m q_i(\zeta)f_i\big(x(\sigma_i(\zeta))\big)\,d\zeta\,.
\]
Since $\lim_{t\to\infty}R(t)=\infty$, there exists $t_2\geq t_1$, such that
$R(t)-R(t_1)\geq\frac{1}{2} R(t)$ for $t\geq t_2$. Then
\beq \label{lowerz}
x(t)\geq \frac12 R(t)w^{1/\gamma}(t)\,.
\eeq
Computing the derivative of $w$, we have
\begin{gather*}
w'(t)= -\sum_{i=1}^m q_i(t)f_i\big(x(\sigma_i(t))\big).
\end{gather*}
Thus $w$ is non-negative and non-increasing. Since $x>0$, by
 (A3),  $f_i\big(x(\sigma_i(t))\big)>0$, and by (A2), it follows that
$\sum_{i=1}^m q_i(t)f_i\big(x(\sigma_i(t))\big)$ cannot be identically zero
in any interval $[b,\infty)$; thus $w'$ cannot be identically zero, and $w$
can not be constant on any interval $[b,\infty)$.
Therefore $w(t)>0$ for $t\geq t_1$.
Computing the derivative,
\beq \label{e28}
\big(w^{1-\beta/\gamma}(t)\big)'
=\Big(1-\frac{\beta}{\gamma}\Big) w^{-\beta/\gamma}(t)w'(t)\,.
\eeq
Integrating \eqref{e28} from $t_2$ to $t$, and using that $w> 0$,
we have
\beq \label{e29}
\begin{aligned}
w^{1-\beta/\gamma}(t_2)
&\geq \Big(1-\frac{\beta}{\gamma}\Big)
\Big[-\int_{t_2}^t w^{-\beta/\gamma}(\eta) w'(\eta)\,d\eta\Big]\\
&= \Big(1-\frac{\beta}{\gamma}\Big)
\Big[\int_{t_2}^t w^{-\beta/\gamma}(\eta)
\Big(\sum_{i=1}^m q_i(\eta)f_i\big(x(\sigma_i(\eta))\big)\Big)\,d\eta\Big]\,.
\end{aligned}
\eeq

Next we find a lower bound for the right-hand side of \eqref{e29},
independent of the solution $x$. By (A3), \eqref{e21a}, \eqref{f/decr}, and \eqref{lowerz}, we have
\begin{align*}
f_i(x(t))
&= f_i\big(x(t)\big)\frac{x^\beta(t)}{x^\beta(t)}
\geq \frac{f_i\big(\delta R(t)\big)}{(\delta R(t))^\beta} x^\beta(t)\\
&\geq \frac{f_i\big(\delta R(t)\big)}{(\delta R(t))^\beta}
\Big(\frac{R(t) w^{1/\gamma}(t)}{2}\Big)^\beta
= \frac{f_i\big(\delta R(t)\big)}{(2\delta)^\beta}
w^{\beta/\gamma}(t)\quad \text{for } t\geq t_2\,.
\end{align*}
Since $w$ is non-increasing, $\beta/\gamma>0$,  and $\sigma_i(\eta)<\eta$, it follows that
\beq \label{e30}
f_i\big(x(\sigma_i(\eta))\big)
\geq \frac{f_i\big(\delta R(\sigma_i(\eta))\big)}
{(2\delta)^\beta}  w^{\beta/\gamma}(\sigma_i(\eta))
\geq \frac{f_i\big(\delta R(\sigma_i(\eta))\big)}
{(2\delta)^\beta}  w^{\beta/\gamma}(\eta)\,.
\eeq
Going back to \eqref{e29}, we have
\beq \label{e32}
\begin{aligned}
w^{1-\beta/\gamma}(t_2)
&\geq \frac{\big(1-\frac{\beta}{\gamma}\big)}{(2\delta)^\beta}
\Big[\int_{t_2}^t \sum_{i=1}^m q_i(\eta)f_i(\delta R(\sigma_i(\eta)))\,d\eta\Big]\,.
\end{aligned}
\eeq
Since $(1-\beta/\gamma)>0$, by \eqref{e24}
the right-hand side approaches $\infty$ as $t\to\infty$.
This contradicts \eqref{e32}  and
completes the proof of sufficiency for eventually positive solutions.

For an eventually negative solution $x$, we introduce the  variables
$y=-x$ and $g_i(y)=-f_i(y)$. Then $y$ is an eventually positive solution
of  \eqref{e1} with $g_i$ instead of $f_i$.
Note that $g_i$ satisfies (A3) and \eqref{f/decr} so can apply the above
process for the solution $y$.

Next we show the necessity part by a contrapositive argument.
When \eqref{e24} does not hold we find a eventually positive solution
that does not converge to zero.
If \eqref{e24} does not hold for some $\delta>0$, then for
each $\epsilon>0$ there exists $t_1\geq t_0$ such that
\beq \label{e34}
\int_\eta^\infty \sum_{i=1}^m q_i(\zeta)f_i(\delta R(\sigma_i(\zeta)))\,d\zeta\leq \epsilon/2
\eeq
for all $\eta \geq t_1$. Note that  $t_1$ depends on $\delta$.
We define the  set of continuous functions
\[
M=\{x\in C([0,\infty)): (\epsilon/2)^{1/\gamma}\big(R(t)-R(t_1)\big)
\leq x(t)
\leq  \epsilon^{1/\gamma} \big(R(t)-R(t_1)\big),\; t\geq t_1\}\,.
\]
We define an operator $\Phi$ on $M$ by
\[
(\Phi x)(t) =\begin{cases}
0  & \text{if }  t\leq t_1\\
 \int_{t_1}^t
\Big[\frac{1}{r(\eta)}\Big[\epsilon/2+ \int_\eta^\infty \sum_{i=1}^m q_i(\zeta)f_i(x(\sigma_i(\zeta)))\,d\zeta)\Big]\Big]^{1/\gamma}\,d\eta
&\text{if } t>t_1\,.
\end{cases}
\]
Note that when $x$ is continuous, $\Phi x$ is also continuous
on $[0,\infty)$.
If $x$ is a fixed point of $\Phi$, i.e.\ $\Phi x=x$, then $x$ is a solution
of \eqref{e1}.

First we estimate $(\Phi x)(t)$ from below. For $x\in M$,
we have $0\leq \epsilon^{1/\gamma}\big(R(t)-R(t_1)\big) \leq x(t)$.
By (A3), we have $0\leq f_i(x(\sigma_i(\eta)))$ and by (A2) we have
\[
(\Phi x)(t)\geq 0+  \int_{t_1}^t
\Big[\frac{1}{r(\eta)}[\epsilon/2 +0 +0 ]\Big]^{1/\gamma}\,d\eta
=(\epsilon/2)^{1/\gamma}\big(R(t)-R(t_1)\big)\,.
\]

Now we estimate $(\Phi x)(t)$ from above.
For $x$ in $M$, by (A2) and (A3), we have
$f_i(x(\sigma_i(\zeta)))\leq f_i\big(\delta R(\sigma_i(\zeta))\big)$.
Then by \eqref{e34},
\begin{align*}
(\Phi x)(t)
&\leq \int_{t_1}^t \Big[\frac{1}{r(\eta)}\Big[\epsilon/2
+\int_\eta^\infty \sum_{i=1}^m q_i(\zeta)f_i(\delta R(\sigma_i(\zeta)))\,d\zeta\Big]\Big]^{1/\gamma}\,d\eta \\
&\leq  \epsilon^{1/\gamma} \big(R(t)-R(t_1)\big)\,.
\end{align*}
Therefore, $\Phi$ maps $M$ to $M$.

Next we find a fixed point for $\Phi$ in $M$.
Let us define a sequence of  functions in $M$ by the recurrence relation
\begin{gather*}
u_0(t)=0 \quad \text{for } t\geq t_0,\\
u_1(t)=(\Phi u_0)(t)=\begin{cases}
0 &\text{if } t< t_1\\
 \epsilon^{1/\gamma} \big(R(t)-R(t_1)\big) &\text{if } t\geq t_1
\end{cases}, \\
u_{n+1}(t) = (\Phi u_n)(t)\quad\text{for }n\geq 1,\; t\geq t_1\,.
\end{gather*}
Note that for each fixed $t$, we have $u_1(t)\geq u_0(t)$. Using that $f$ is
non-decreasing and mathematical induction, we can show that
$u_{n+1}(t)\geq u_n(t)$. Therefore, the sequence $\{u_n\}$
converges pointwise to a function $u$.
Using the Lebesgue Dominated Convergence Theorem, we can show  that
$u$  is a fixed point of $\Phi$ in $M$.
This shows under assumption \eqref{e34}, there a non-oscillatory solution
that does not converge to zero. This completes the proof.
\end{proof}

In the next theorem, we assume the  existence of  a differentiable
function $\sigma_0$ such that
\beq \label{sigma0}
0<\sigma_0(t)\leq \sigma_i(t),\quad
\exists \alpha>0 : \alpha\leq \sigma_0'(t),\quad
\text{for } t\geq t_0,\; i=1,2,\dots, m.
\eeq
Also we assume that there exists a constant $\beta$,
the quotient of two positive odd integers, with $\gamma<\beta$, such that
\beq \label{f/incr}
\frac{f_i(u)}{u^\beta}\text{ is non-decreasing for }0<u,\; i=1,2,\dots, m\,.
\eeq
For example  $f_i(u)=|u|^\alpha\operatorname{sgn}(u)$, with $\beta<\alpha$
satisfies this condition.

\begin{theorem} \label{thm2}
Under assumptions {\rm (A1)--(A4)}, \eqref{sigma0},  \eqref{f/incr}, and $r(t)$ is non-decreasing, every solution of \eqref{e1} is oscillatory if and only if
\beq \label{e42}
\int_{t_1}^\infty\Big[ \frac{1}{r(\eta)}\int_\eta^\infty \sum_{i=1}^m q_i(\zeta)\,d\zeta \Big]^{1/\gamma}\,d\eta
=\infty\,.
\eeq
\end{theorem}

\begin{proof}
We prove sufficiency by contradiction. Initially assume that $x$ is an
eventually positive solution that does not converge to zero.
Using the same argument as in  Lemma \ref{lem1},
there exists $t_1\geq t_0$ such that:
$x(\sigma_i(t))>0$, $x(\tau(t))>0$, and $r(t)\big(x'(t)\big)^\gamma$ is positive and
non-increasing. Since $r(t)>0$ so  $x(t)$ is increasing for $t\geq t_1$. From (A3), $x(t)\geq x(t_1)$ and \eqref{f/incr}, we have
\[
f_i(x(t)) \geq \frac{f_i(x(t))}{x^\beta(t)}x^\beta(t)
\geq \frac{f_i(x(t_1))}{x^\beta(t_1)}x^\beta(t)\,.
\]
By (A1) there exists a $t_2\geq t_1$ such that $\sigma_i(t)\geq t_1$
for $t\geq t_2$.
Then
\beq \label{e44}
f_i\big(x(\sigma_i(t))\big) \geq \frac{f_i(x(t_1))}{x^\beta(t_1)}x^\beta
(\sigma_i(t))\quad \forall t\geq t_2\,.
\eeq
Using this inequality, \eqref{e23}, that $\sigma_i(t)\geq \sigma_0(t)$
 which is an increasing function,
and that $z$ is increasing, we have
\[
r(t)\big(x'(t)\big)^\gamma
\geq \frac{x^\beta(\sigma_0(t))}{x^\beta(t_1)}\int_t^\infty  \sum_{i=1}^m q_i(\eta) f_i(x(t_1))\,d\eta\,,
\]
for $t\geq t_2$.
From $r(t)\big(z'(t)\big)^\gamma$ being non-increasing and $\sigma_0(t)\leq t$,
we have
\[
r(\sigma_0(t))\big(x'(\sigma_0(t))\big)^\gamma
\geq r(t)\big(x'(t)\big)^\gamma .
\]
We use this in the left-hand side of the above inequality.
Then dividing by $r(\sigma_0(t))>0$, raising both sides to
the $1/\gamma$ power, and dividing by
$z^{\beta/\gamma}(\sigma_0(t))>0$, we have
\begin{align*}
\frac{x'(\sigma_0((t))}{x^{\beta/\gamma}(\sigma_0(t))}
\geq \Big[ \frac{1}{r(\sigma_0(t))x^\beta(t_1)}\int_t^\infty  \sum_{i=1}^m q_i(\eta) f_i(x(t_1)) \,d\eta\Big]^{1/\gamma}\,,
\end{align*}
for $t\geq t_2$.
Multiplying the left-hand side by $\sigma_0'(t)/\alpha \geq 1$,
and integrating from $t_1$ to $t$,
\beq \label{e45}
\begin{aligned}
\frac{1}{\alpha} \int_{t_1}^t
\frac{z'(\sigma_0(\eta)) \sigma_0'(\eta)}{z^{\beta/\gamma}(\sigma_0(\eta))}\,d\eta
&\geq \frac{1}{z^{\beta/\gamma}(t_1)}
\int_{t_1}^t \Big[\frac{1}{r(\sigma_0(\eta))}\int_\eta^\infty  \sum_{i=1}^m q_i(\zeta)f_i(x(t_1)) \,d\zeta\Big]^{1/\gamma}\,d\eta.
\end{aligned}
\eeq
On the left-hand side, since $\gamma<\beta$, integrating, we have
\[
\frac{1}{\alpha(1-\beta/\gamma)}\Big[z^{1-\beta/\gamma}(\sigma_0(\eta))\Big]_{s=t_2}^t
\leq \frac{1}{\alpha(\beta/\gamma-1)} z^{1-\beta/\gamma}(\sigma_0(t_2)) \,.
\]
On the right-hand side of \eqref{e45}, we use that
$\min_{1\leq i\leq m} f_i(z(t_1))>0$ and that $r(\sigma_0(s))\leq r(s)$,
to conclude that \eqref{e42} implies the right-hand side approaching $\infty$,
as $t\to\infty$. This contradiction
implies that the solution $x$ cannot be eventually positive.

For eventually negative solutions, we use the same change of variables as in
Theorem \ref{thm1}, and proceed as above.

To prove the necessity part we assume that \eqref{e42} does not hold, and
obtain an eventually positive solution that does not converge to zero.
If \eqref{e42} does not hold, then for each $\epsilon>0$ there exists $t_1\geq t_0$
such that
\beq \label{e42n}
\int_{t_1}^\infty\Big[ \frac{1}{r(\eta)}\int_\eta^\infty \sum_{i=1}^m q_i(\zeta)\,d\zeta\Big]^{1/\gamma}\,d\eta
<\epsilon/2(f_i(\epsilon))^{1/\gamma} \quad \forall t\geq t_1\,.
\eeq
Let us consider the set of continuous functions
\[
M=\{x\in C([0,\infty)):\epsilon/2\leq x(t) \leq \epsilon\text{ for }t\geq t_1\}
\]
Then we define the operator
\[
(\Phi x)(t)=\begin{cases}
0 &\text{if } t\leq t_1,\\[3pt]
\epsilon/2
+ \int_{t_1}^t\frac{1}{r(\eta)}\Big[\int_\eta^\infty \sum_{i=1}^m q_i(\zeta)f_i(x(\sigma_i(\zeta)))\,d\zeta\Big]^{1/\gamma}\,d\eta
&\text{if } t>t_1\,.
\end{cases}
\]
Note that if $x$ is continuous, $\Phi x$ is also continuous at $t=t_1$. Also note that if $\Phi x=x$, then $x$ is solution of \eqref{e1}.

First we estimate $(\Phi x)(t)$ from below. Let $x\in M$. By $0<\epsilon/2 \leq x$, we have $(\Phi x)(t)\geq \epsilon/2+0+0$, on $[t_1,\infty)$.

Now we estimate $(\Phi x)(t)$ from above.  Let $x\in M$. Then $x\leq \epsilon$ and by \eqref{e42n} , we have
\begin{align*}
(\Phi x)(t)
&\leq \epsilon/2+\big(f_i(\epsilon)\big)^{1/\gamma}
\int_{t_1}^t \Big[\frac{1}{r(\eta)}\int_\eta^\infty \sum_{i=1}^m q_i(\zeta)\,d\zeta
\Big]^{1/\gamma}\,d\eta \leq \epsilon/2+\epsilon/2=\epsilon.
\end{align*}
Therefore $\Phi$ maps $M$ to $M$. To find a fixed point for $\Phi$ in $M$, we define a sequence of functions by
the recurrence relation
\begin{gather*}
u_0(t)=0\quad \text{for } t\geq t_0,\\
u_1(t)=(\Phi u_0)(t)= 1 \quad\text{for }t\geq t_1,\\
u_{n+1}(t)=(\Phi u_n)(t)\quad \text{for } n\geq 1,\; t\geq t_1\,.
\end{gather*}
Note that for each fixed $t$, we have $u_1(t)\geq u_0(t)$. Using that $f$ is non-decreasing
and mathematical induction, we can prove that $u_{n+1}(t)\geq u_n(t)$.
Therefore $\{u_n\}$ converges pointwise to a function $u$ in $M$.
Then $u$ is a fixed point of $\Phi$ and a positive solution to \eqref{e1}. This completes the proof.
\end{proof}



\begin{example} \rm
Consider the delay differential equations
\begin{gather} \label{exam1}
\Big(e^{-t}\big(x'(t)\big)^{11/3}\Big)'
+\frac{1}{t+1}(x(t-2))^{1/3}
+ \frac{1}{t+2}(x(t-1))^{5/3}=0\,.
\end{gather}
Here $\gamma = 11/3$, $r(t)=e^{-t}$, $\sigma_1(t)=t-2$,
$\sigma_2(t)=t-1$, $R(t)=\int_0^t e^{5s/3}\,ds= \frac{3}{5}\big(e^{5t/3}-1\big)$, $f_1(x)=x^{1/3}$
and $f_2(x)=x^{5/3}$. For $\beta=7/3$, we have
$0<\max\{\alpha_1,\alpha_2\}<\beta<\gamma$, and
$f_1(x)/x^\beta=x^{-2}$ and $f_2(x)/x^\beta=x^{-2/3}$ which both are decreasing
functions.
To check \eqref{e24} we have
\begin{align*}
\int_0^\infty \sum_{i=1}^m  &q_i(\eta)f_i(\delta R(\sigma_i(\eta)))\,d\eta \geq  \int_0^\infty \sum_{i=1}^m
 q_i(\eta)f_i(\delta R(\sigma_i(\eta)))\,d\eta  \\
& \geq \int_0^\infty q_1(\eta)f_1(\delta R(\sigma_1(\eta)))\,d\eta \\
&=\int_0^{\infty}\frac{1}{\eta+1} \Big(\delta\frac{3}{5}
\big(e^{5(\eta-2)/3}-1\big)\Big)^{1/3}\,d\eta =\infty \quad\forall \delta>0,
\end{align*}
since the integral approaches $+\infty$ as $\eta \to +\infty$.
So, all the conditions of Theorem \ref{thm1} hold, and therefore, each solution
of \eqref{exam1} is oscillatory or converges to zero.
\end{example}

\begin{example} \rm
Consider the neutral differential equations
\begin{gather} \label{exam3}
\Big(\big(x'(t)\big)^{1/3}\Big)'+t(x(t-2))^{7/3}
+ (t+1)(x(t-1))^{11/3}=0 \,.
\end{gather}
Here $\gamma = 1/3$, $r(t)=1$, $\sigma_1(t)=t-2$, $\sigma_2(t)=t-1$, $f_1(v)=v^{7/3}$ and $f_2(v)=v^{11/3}$.
For $\beta=5/3$, we have $\min\{\alpha_1,\alpha_2\}>\beta>\gamma$, and
$f_1(x)/x^\beta=x^{2/3}$ and $f_2(x)/x^\beta=x^{2}$ which both are increasing
functions. To check \eqref{e42} we have
\begin{align*}
 \int_{t_0}^\infty\Big[ \frac{1}{r(\eta)}
\int_\eta^\infty \sum_{i=1}^m q_i(\zeta)\,d\zeta\Big]^{1/\gamma}\,d\eta  & \geq \int_{t_0}^\infty\Big[ \frac{1}{r(\eta)}
\int_\eta^\infty \sum_{i=1}^m q_i(\zeta)\,d\zeta \Big]^{1/\gamma}\,d\eta \\
& \geq \int_{t_0}^\infty\Big[ \frac{1}{r(\eta)}\int_\eta^\infty q_1(\zeta)\,d\zeta \Big]^{1/\gamma}\,d\eta  \\
&\geq \int_2^{\infty} \Big[\int_\eta^\infty \zeta\, d\zeta\Big]^{3}\,d\eta=\infty.
\end{align*}
So, all the conditions of of Theorem \ref{thm2} hold. Thus, each solution of
\eqref{exam3} is oscillatory or converges to zero.
\end{example}

\section*{Open Problem}
Based on this work and \cite{Baculikova2,Baculikova3,boh,dzurina,Grace,karpuz,li1,li2,santra1,akt}
an open problem that arises is to establish necessary and sufficient conditions
for the oscillation of the solutions of the second-order nonlinear neutral
differential equation
\begin{equation}
\Big(r(t)\big(z'(t)\big)^\gamma\Big)' +\sum_{i=1}^m q_i(t)f_i\big(x(\sigma_i(t))\big)=0, \text{ for } t \geq t_0,\notag
\end{equation}
where $z(t)=x(t)+p(t)x(\tau(t))$ for $p\in C(\mathbf{R},\mathbf{R})$.

%%%


\end{document}
